Author's response to reviews

Title: Highly Active Antiretroviral Therapy and Dyslipidemia in people living with HIV/AIDS in Fako Division, South West Region of Cameroon

Authors:

Dickson Shey Nsagha (nsaghads@hotmail.com)
Elroy Patrick Weledji (elroypat@yahoo.co.uk)
Nguedia Jules Clement Assob (julesclement@yahoo.fr)
Anna Longdoh Njunda (ann_njunda@yahoo.com)
Elvis Asangbeng Tanue (asangbengelvis@gmail.com)
Odette Dzemo Kibu (yayaodette@yahoo.com)
Charlotte Wenze Ayima (ayimacharlotte@yahoo.com)
Marcelin Ngowe Ngowe (nkouki2002@yahoo.fr)

Version: 2
Date: 6 August 2015

Author's response to reviews:

resubmission cover letter has been attached